Author's response to reviews

Title: Stimulated monocyte IL-6 secretion predicts survival of patients with head and neck squamous cell carcinoma

Authors:

John-Helge Heimdal (john.heimdal@helse-bergen.no)
Kenneth Kross (kenneth.kross@kir.uib.no)
Beate Klementsen (beate.klementsen@abm.uib.no)
Jan Olofsson (jan.olofsson@helse-bergen.no)
Hans Jorgen Aarstad (hans.jorgen.aarstad@helse-bergen.no)

Version: 7 Date: 10 January 2008

Author's response to reviews: see over

Dear Editor!

Please find enclosed the final version of the manuscript "Stimulated monocyte IL-6 secretion predicts survival of patients with head and neck squamous cell carcinoma".

We have now written the author list and reference list as requested. All references are cited in the text using consecutive numbers in square brackets. Figure titles are erased and information about the figures is specified in the manuscript (in the figure legends) positioned after the reference list. Additional files (title page, abstract, and tables) are deleted from the submission system.

Yours sincerely,

John-Helge Heimdal, MD, PhD
Staff Surgeon
Dept of ORL/Head and Neck Surgery
Haukeland University Hospital